Author's response to reviews

Title: Targeting of human interleukin-12B by small hairpin RNAs in xenografted psoriatic skin

Authors:

Rasmus O Bak (rasmusbak@humgen.au.dk)
Karin Stenderup (karin.stenderup@ki.au.dk)
Cecilia Rosada (cecilia.kjeldsen@ki.au.dk)
Line B Petersen (line.barretti@studmed.au.dk)
Brian Moldt (bmoldt@scripps.edu)
Frederik Dagnæs-Hansen (mikrfdh@medmicro.au.dk)
Maria Jakobsen (maria@humgen.au.dk)
Søren Kamp (soka@regionsjaelland.dk)
Thomas G Jensen (thomas@humgen.au.dk)
Tomas N Dam (TNDam@dadlnet.dk)
Jacob G Mikkelsen (giehm@humgen.au.dk)

Version: 3 Date: 14 February 2011

Author's response to reviews:

The required major and minor formatting changes have been made:

- Author list
- Authors' contributions: all authors read and approved the final manuscript
- Removal of coloured text
- Removal of visible vertical lines from the table
- Cropping of figures and removal of figure number and title